Author’s response to reviews

Title: The Impact of Gender and Academic Achievement on the Violation of Academic Integrity for Medical Faculty Students, a descriptive cross-sectional survey study

Authors:
Müesser Özcan (muglabiyomer@gmail.com)
Emine Nese Yeniceri (nese_yeniceri@yahoo.com)
Edip Güvenc Cekic (edipgcekic@mu.edu.tr)

Version: 4 Date: 16 Oct 2019

Author’s response to reviews:

Dear Editor and Reviewers,

On behalf of all authors I would like to thank the editor and reviewers for their valuable comments and interest in our paper.

Nourhan Aly (Reviewer 4): I would like to thank the authors for responding to all comments.

Minor comments:
1. I think the author should have used the 2 previous studies conducted in Turkey (on nursing students) in the discussion part (whether their findings agree with yours or not).
   - The results of these studies have been included in the discussion part.

2. Sixth year students had higher tendency towards academic dishonesty when compared to all other years and the multiple comparison revealed statistically significant difference between 4th and 5th years when compared to 6th year students. Can you explain this finding? (your own assumption about why this occurred?)